Author's response to reviews

Title: Fruit and vegetables intake among elderly Iranians: a theory-based interventional study using the Five-A-Day program

Authors:

Leili Salehi (l.salehi@razi.tums.ac.ir)
Kazem Mohammad (mohamadk@tums.ac.ir)
Ali Montazeri (montazeri@acecr.ac.ir)

Version: 5 Date: 12 November 2011

Author's response to reviews:

Emma Mullen
Editorial Production Assistant
BioMed Central
Liverpool Science Park
131 Mount Pleasant
Liverpool
L3 5TF

Dear Emma,

Thank you for your recent e-mail. Now tables are renumbered and they are cited in numerical order.

Best wishes.
Ali Montazeri